Author's response to reviews

Title: Duodenal Somatostatinoma: A Case Report and Review of Literature

Authors:

oguz koc (oguzcoach@yahoo.com)
yigit duzkoylu (dryigit@gmail.com)
yavuz s sari (yssari2003@yahoo.com)
hasan bektas (hasanbektas3417@hotmail.com)
gungor uzum (guzum@yahoo.com)
vahit tunali (vtunali@hotmail.com)
evra dursun (durnevra@gmail.com)

Version: 6 Date: 26 February 2013

Author's response to reviews: see over
Dear Editor,

We described the features of the tumor more precisely, by the way consulted with the pathology department once more (and changed one of the names between the authors), obtained a picture of the specimen and deleted the word “psammoma bodies” from the manuscript.

Yours sincerely,
Dr Oguz Koc
Dr Yigit Duzkoaylu